Author’s response to reviews

Title: Severe colonic bleeding in ulcerative colitis is refractory to selective transcatheater arterial embolization

Authors:

Jose Miranda-Bautista (pepon_miranda@hotmail.com)
Lucía Diéguez (lucia.dieguez@salud.madrid.org)
Gracia Rodríguez-Rosales (gracia.rodriguez@salud.madrid.org)
Ignacio Marín-Jiménez (drnachomarin@hotmail.com)
Luis Menchén (lamenchengugm@salud.madrid.org)

Version: 2 Date: 15 Mar 2019

Author’s response to reviews:

Editor comments

1. The entire manuscript has been edited by an English editing, on-line service. Several grammar, spelling and punctuation issues have been corrected throughout the manuscript.

2. Corrected, as suggested

3. Corrected, as suggested

4. Corrected, as suggested (Hispanic)

5. Corrected, as suggested

6. Added, as suggested

7. Corrected, as suggested

8. Corrected, as suggested

9. Corrected, as suggested

Reviewer #2 comments

1. Corrected, as suggested
2. Corrected, as suggested